Author’s response to reviews

Title: Factors associated with knowledge of hypertension among adolescents: implication for preventive education programs in primary care.

Authors:

Iga Grad (igagrad@o2.pl)
Agnieszka Mastalerz-Migas (agnieszka.migas@gmail.com)
Katarzyna Kilis-Pstrusinska (katarzyna.kilis-pstrusinska@umed.wroc.pl)

Version: 2 Date: 27 November 2014

Author’s response to reviews:

No changes were made in the cover letter.